Author’s response to reviews

Title: Italian Version of Nursing Students' Perception of Instructor Caring (I-NSPIC): assessment of Reliability and Validity

Authors:

Cristina Arrigoni (cristina.arrigoni@unipv.it)
Mariangela Puci (mariangela.puci@unipv.it)
Anna Maria Grugnetti (annamaria.grugnetti@unipv.it)
Lucia Collivasone (lucia.collivasone@unipv.it)
Elisa Fenizia (e.fenizia@unicampus.it)
Paola Borrelli (paola.borrelli@unipv.it)
Ercole Vellone (ercole.vellone@uniroma2.it)
Rosaria Alvaro (rosaria.alvaro@uniroma2.it)
Michela Piredda (michela.piredda@unicampus.it)
Maria Grazia De Marinis (m.demarinis@unicampus.it)

Version: 2 Date: 21 Sep 2017

Author’s response to reviews:

MEED-D-17-00146R1

Italian Version of Nursing Students' Perception of Instructor Caring (I-NSPIC): assessment of Reliability and Validity

Cristina Arrigoni, Research Scientist in Nursing Science; Mariangela Puci, Research Fellow in Biostatistics and Epidemiology; Anna Maria Grugnetti, Ph.D. MNSc, RNA. Adjunct Professor in Nursing Scie; Lucia Collivasone, Preceptor of Nursing Degree Program.; Elisa Fenizia, Ph.D Student in Nursing Science; Paola Borrelli, Ph.D Research Scientist in Biostatistics and Epid; Ercole Vellone, Ph.D in Nursing Science. Assistant Professor; Rosaria Alvaro, MNSc Associate
Dear Editor

We have completed the submission process:

We uploaded a single final, clean version of the manuscript with tracked changes switched off and we uploaded the 5 tables mentioned in the manuscript text with related legends at the end of the manuscript. We added a section “Supplementary files” and we inserted additional files (supplementary materials) explicitly referred to in the main text.

We removed from the file inventory any files we do not wish to see published alongside our manuscript, such as cover letter, rebuttal letter, manuscript with tracked changes, etc.

Best wishes

Anna Maria Grugnetti